Author's response to reviews

**Title:** Ethical issues relating to pre-diabetic living donor kidney transplantation

**Authors:**

Aldo Ferreira-Hermosillo (aldo.nagisa@gmail.com)
Edith Valdez-Martínez (edith.valdez@imss.gob.mx)

**Version:** 2  **Date:** 5 September 2013

**Author's response to reviews:** see over
Dear Dr. Aldcroft,

Please find attached the revised manuscript entitled **ETHICAL ISSUES RELATING TO PRE-DIABETIC LIVING DONOR KIDNEY TRANSPLANTATION** by Ferreira-Hermosillo A and Valdez-Martínez E. This manuscript is submitted for possible publication in the BMC Medical Ethics journal. According to initial reviewer’s recommendation we seek for assistance of a fluent English speaking colleague who already corrected our language with particular attention to the abstract. We hope that with these changes you can proceed with peer review.

Kind regards.

MD. Aldo Ferreira-Hermosillo
Endocrinologist